Author's response to reviews

Title: Quality of care for older adults with chronic obstructive pulmonary disease and asthma based on comparisons to practice guidelines and smoking status

Authors:

Benjamin M Craig (ben.craig@fammed.wisc.edu)
Connie K Kraus (ckkraus@pharmacy.wisc.edu)
Betty A Chewning (bachelwning@pharmacy.wisc.edu)
James E Davis (Jim.Davis@fammed.wisc.edu)

Version: 2 Date: 14 November 2007

Author's response to reviews:

Please find the enclosed manuscript entitled “Quality of Care for Older Adults with Chronic Obstructive Pulmonary Disease and Asthma Based on Comparisons to Practice Guidelines and Smoking Status” for submission to BMC Health Services Research as an Original Research manuscript.

I can attest that this manuscript has not been published and is not being considered for print by any other publication and that the authors have no conflicts of interest.

Should you have any questions or require anything further, please do not hesitate to contact me at the address included.

Sincerely,
Benjamin M. Craig, PhD